Author’s response to reviews

Title: A mutation screening of oncogenes and nuclear encoded mitochondrial complex I genes in oncocytic thyroid tumors

Authors:

Cecilia Evangelisti (cecilia.evangelisti2@unibo.it)
Dario de Biase (dario.debiase@unibo.it)
Ivana Kurelac (ivana_kurelac@yahoo.com)
Claudio Ceccarelli (claudio.ceccarelli@unibo.it)
Holger Prokish (Prokisch@helmholtz-muenchen.de)
Thomas Meitinger (meitinger@helmholtz-muenchen.de)
Paola Caria (paola.caria@unica.it)
Roberta Vanni (paola.caria@unica.it)
Giovanni Romeo (egf.giovanni.romeo@gmail.com)
Giovanni Tallini (giovanni.tallini@unibo.it)
Giuseppe Gasparre (giuseppe.gasparre@gmail.com)
Elena Bonora (elena.bonora6@unibo.it)

Version: 2
Date: 24 September 2014

Author’s response to reviews:

To the Editorial Board of BMC Cancer,
we made changes required by the Journal Editorial Office. In particular:
1) we introduced line numbering
2) we included e-mail addresses of all authors on the title page
3) we added the name of the ethics committee (see line 153-154, page 7)
4) we specified that all patients gave their approvals (see line 155-156, page 7)